\begin{table}%t1
\caption{\label{tab:samp}(1) Carignan \& Beaulieu (\cite{Carignan1989}); (2) Broeils (\cite{Broeils1992a}); (3) Sanders (\cite{Sanders1996}); (4) Lake et~al. (\cite{Lake1990}); (5) Martimbeau et~al. (\cite{Martimbeau1994}); (6)~Newton (\cite{Newton1980}); (7) Corbelli (\cite{Corbelli2003}); (8) Broeils (\cite{Broeils1992b}); (9) Wevers (\cite{Wevers1984}); (10) Begeman (\cite{Begeman1987}); (11) Kent (\cite{Kent1987}); (12) Sicking (\cite{Sicking1997}); (13) Chemin et~al. (\cite{Chemin2006}); (14) Carignan \& Puche (\cite{Carignan1990b}); (15) Carignan (\cite{Carignan1985}); (16) Puche et~al. (\cite{Puche1991}); (17) Meurer et~al. (\cite{Meurer1994}); (18) Kent (\cite{Kent1986}); (19)~Puche et~al. (\cite{Puche1990}); (20) Begeman (\cite{Begeman1989}); (21) Puche et~al. (1991a); (22) Broeils \& Knapen (\cite{Broeils1991}); (23) Kent (\cite{Kent1984}); (24) Cote et~al.(\cite{Cote1991}); (25) Sancisi \& van Albada (\cite{Sancisi1987}); (26) Carignan et~al. (\cite{Carignan1990}); (27) Carignan \& Puche (\cite{Carignan1990a}); (28) de Blok et~al. (1985); (29) Carignan et~al. (\cite{Carignan1988}); (30) Roelfsema \& Allen (\cite{Roelfsema1985}); (31) Leroy et~al.  (\cite{Leroy2005}).}
\small%\centering
\par
\begin{tabular}{lrcclrcccl}
\hline
\hline
Name & $D$        & $L_{\rm B}$             & $M_{\rm HI}$             &Type& $V_{\rm max}$      & Inclination &  $r_{\star}$ &  $r_{\rm max}/r_{\star}$  &References  \\
           & (Mpc)  &  ($L_{\odot}$)&  ($M_{\odot}$)     &          &   (km~s$^{-1}$)&     (degree)&    (kpc)         &                                &              \\
\hline
DDO154	&   4.0	&   3.4E07  &  1.8E08      			&I&  43 	& 32 &   0.5   &    15   & (1)            \\
DDO168	&   4.0	&   2.2E08  &  1.8E08       			&I&  52 	& 80 &   0.5   &      8   & (2)(3)         \\
DDO170	&  15.8	&   3.1E07  &  1.7E09       			&I&  66 	& 80 &   0.6   &    19   & (4)            \\
IC~2574	&   2.5 	&   5.4E08  &  2.1E08    			&SABm&  67 	& 80 &   1.1   &      6   & (5)  (31)          \\
M~33	&   0.8 	&   2.8E09  &  1.8E09      			&Sc& 134 	& 55 &   1.7   &      9   & (6)(7)        \\
NGC~1003	&  11.8 	&   5.7E09  &  2.5E09     			&Sc& 112 	& 81 &   2.6   &    11   & (2)(31)            \\
NGC~1560	&   3.0 	&   6.6E08  &  7.0E08     		&Scd&  78 	& 83 &   0.9   &      8   & (8)            \\
NGC~2366	&   2.6 	&   3.4E08  &  3.4E08    		&IB&  45 	& 90 &   0.8   &      7   & (9)            \\
NGC~2403	&   3.6 	&   5.9E09  &  3.6E09    			&SABc& 140& 60 &   2.9   &      6   & (10)(11)(12)   \\
NGC~24	&   6.8 	&   1.1E09  &  3.8E08     			&Sc& 109     & 70 &   1.1   &      9   & (13)           \\
NGC~247	&   2.8 	&   2.2E09  &  8.0E08    			&SABc& 107     & 75 &   1.9   &      5   & (14) (15)      \\
NGC~253	&   2.6 	&   1.2E10  &  6.6E08    			&SABc& 224 	& 78 &   2.1   &      4   & (16)         \\
NGC~2841	&  20.0 	&   4.9E10  &  2.1E10     			&Sb& 292 	& 68 &   3.3   &    26   & (10)(11)       \\
NGC~2903	&  10.4 	&   2.5E10  &  5.8E09    			&SABb& 181 	& 56 &   2.9   &    13   & (10)(11)       \\
NGC~2915	&   5.6 	&   4.7E08  &  7.7E08     			&SBa&  93 	& 63 &   0.4   &    42   & (17)           \\
NGC~2998	&  70.7 	&   3.9E10  &  2.8E10    			&SABc& 200 	& 61 &   5.0   &    10   & (2) (18)       \\
NGC~300	&   2.9 	&   3.5E09  &  1.4E09     			&Scd&  93 	& 39 &   2.6   &      6   & (19)           \\
NGC~3109	&   1.9 	&   9.8E08  &  3.8E08     			&SBm&  67 	& 80 &   1.0   &      9   & (11)          \\
NGC~3198	&  11.6 	&   1.1E10  &  7.3E09     		&Sc& 149 	& 70 &   2.9   &    12   & (11)(20)      \\
NGC~3726	&  14.9 	&   1.3E10  &  7.6E09      	 		&Sc& 160 	& 49 &   3.7   &      8   & (9)           \\
NGC~4203	&  15.1 	&   3.9E09  &  7.5E08   			&E-SO& 152 	& 90 &   2.3   &      7   & (9)           \\
NGC~4242	&   7.3 	&   1.3E09  &  4.5E08      			&Sd&  98 	& 51 &   1.5   &      7   & (9)           \\
NGC~4258	&   7.3 	&   1.7E10  &  3.7E09    			&SABb& 216 	& 72 &   2.5   &    11   & (9)  \\
NGC~4395	&   2.7 	&   6.4E08  &  4.7E08      		&Sm&  85 	& 90 &   1.5   &      4   & (9)  \\
NGC~45	&   5.9 	&   1.2E09  &  1.3E09    			&SABd& 100 	& 60 &   2.0   &      8   & (13) \\
NGC~4725	&  11.9 	&   1.5E10  &  2.6E09    			&SABa& 223 	& 54 &   4.3   &     6    & (9)  \\
NGC~5033	&  18.4 	&   2.5E10  &  1.5E10     			&Sc& 196 	& 66 &   5.5   &     8    & (10)(18)  \\
NGC~5055	&  10.3 	&   2.3E10  &  9.8E09     			&Sbc& 172 	& 56 &   3.8   &   12    & (9)\\
NGC~5371	&  40.0 	&   5.6E10  &  9.9E09     			&Sbc& 313 	& 54 &   7.4   &     5    & (9) (10) \\
NGC~55	&   1.6 	&   4.6E09  &  6.8E08     			&SBm&  86 	& 85 &   1.4   &     7    & (21)  \\
NGC~5533	&  58.3 	&   3.5E10  &  3.0E10     			&Sab& 230 	& 60 &   5.5   &   14    & (2)  (22)  (23)\\
NGC~5585	&   7.6 	&   1.9E09  &  1.2E09    		&SABc&  89 	& 53 &   1.7   &     6    & (24)  \\
NGC~5907	&  11.0 	&   1.3E10  &  2.4E09      			&Sc& 219 	& 87 &   2.4   &   13    & (25)  \\
NGC~6503	&   4.8 	&   2.0E09  &  1.0E09     	 		&Sc& 122 	& 74 &   1.1   &   15    & (2) (9) \\
NGC~6674	&  51.8 	&   3.4E10  &  2.7E10      			&Sb& 240 	& 62 &   5.5   &   13    & (2)  \\
NGC~6946	&   6.7 	&   1.8E10  &  7.8E09    			&SABc& 159 	& 31 &   3.4   &     5    & (26)  \\
NGC~7331	&  12.8 	&   2.8E10  &  7.3E09     			&Sbc& 241 	& 75 &   2.2   &   14    & (2)(11)  \\
NGC~7793	&   4.1 	&   2.8E09  &  1.0E09     			&Scd&  90 	& 53 &   1.5   &     5    & (27)  \\
NGC~801	&  84.0	&   5.1E10  &  1.9E10      			&Sc& 218 	& 85 &   2.9   &   20    & (2) (18) \\
NGC~925	&   6.5 	&   4.1E09  &  2.5E09     			&Scd& 114 	& 61 &   2.2   &     7    & (9) \\
UGC~128	&  56.4 	&   1.0E09  &  6.2E09      			&Sd& 128 	& 32 &   5.0   &     7    & (28)  \\
UGC~2259	&  10.0 	&   2.3E08  &  3.7E08     		&Sbd&  90 	& 53 &   1.1   &     7    & (11) (31) \\
UGC~2885	&  84.0 	&   5.6E10  &  5.9E10     			&Sc& 298 	& 62 &11.5    &     6    & (18)  \\
\hline                                              
\end{tabular}
\par
\end{table}